\begin{table}%TB.1
%\centering
\small
\caption{\label{tbl-binsys}
  Published double-star parameters and spectral types, see also
  Sect.~\ref{sec-binsys} and Appendix~\ref{sec-app-b}.
} %% end-caption
\newcommand{\spect}{\multicolumn{2}{c}{spectral}}
\newcommand{\types}{\multicolumn{2}{c}{types}}
\newcommand{\sep}  {\multicolumn{1}{c}{sep}}
\newcommand{\usep} {\multicolumn{1}{c}{{($''$)}}}
\newcommand{\pa}   {\multicolumn{2}{c}{PA}}
\newcommand{\upa}  {\multicolumn{2}{c}{{(\degr)}}}
\newcommand{\dv}   {\multicolumn{2}{c}{\DV}}
\newcommand{\udv}  {\multicolumn{2}{c}{{(mag)}}}
\newcommand{\nop}  {\multicolumn{2}{c}{--}}
\newcommand{\nix}  {\multicolumn{2}{c}{~~~}}
\newcommand{\apa}  {(ap)}
\newcommand{\per}  {(peri)}
\par
\begin{tabular}{l l@{+}l l r@{.}l r@{.}l l}\hline\hline
 double      &\spect            &\sep  &\pa    &\dv        &          \\
 star         &\types            &\usep &\upa   &\udv       &ref.      \\
\kline
XY~Per        &A2\scii  &A2-5    &1.33  & 76&   &$\sim$0&64 &7      \\
UX~Ori        &A2\scii  &??      &0.022 &257&   &      \nop &5      \\
HD~150193      &A4\scv   &??      &1.09  &225&   &$\sim$3&5  &7, 12   \\
%%%\fnv       &A3       &G5      &--&\nop   &      1&0  &11     \\ %% BD{+24}{676}
BF~Ori        &A5       &??      &--&\nop   &      \nop &5      \\
HR~5999        &A5\sciii &IR      &1.44  &112&   &$\geq$3&65 &9, 10, 13\\
HD~144432      &A9\sciii &K0-5    &1.4   &353&   &      \nop &16     \\
NX Pup        &F2\sciii &G5\sciv &0.126 & 63&   &      0&64 &4, 6, 7  \\
\fnp          &A2-4     &IR      &4.18  & 43&   &  $<$0 &14 &12     \\ %% BD{+46}{3471}
HK~Ori        &A4\scv   &IR      &0.34  & 41&   &$\sim$2&9  &9, 10   \\
AK~Sco \apa   &F5\scv   &F5\scv  &0.001 &185&   &      0&42 &2      \\
AK~Sco \per   &\nix              &      &\nix   &      0&09 &       \\ %% ref sect 5.10
Z~CMa         &F5\sci b &B0      &0.10  &122&5  &      2&1  &3      \\
HD~199143      &F6-8     &K7-M0   &1.078 &325&   &      \nop &1, 15   \\
V380~Ori      &A2\scv   &IR      &0.134 &204&   &$\geq$2&6  &9, 10   \\
KK~Oph        &A6\scv   &IR      &1.61  &247&   &$\sim$1&0  &10, 12  \\
%%%RY~Tau     &F8       &??      &0.024 &304&   &      \nop &5      \\
HD~98922      &A0       &??      &--&\nop   &      \nop &8, 14   \\
%%%T Tau      &K1\sciii &IR      &0.73  &176&   &      \nop &       \\
\kline \end{tabular}
\medskip
\begin{tiny}
\hspace*{-0.3cm}\begin{tabular}{ r@{:~}l r@{:~}l rl }
 1& v.d.Ancker et~al. (\cite{vdA00})   & 7& ESA~(1997)             &13:&\hskip-0.25cm Stecklum et~al. (\cite{Stecklum95}) \\
 2& Andersen et~al. (\cite{Andersen89})     & 8& de~Groot (\cite{Groot71})       &14:&\hskip-0.25cm Bond  (\cite{Bond70})           \\
 3& Barth et~al. (\cite{Barth94})        & 9& Leinert et~al. (\cite{Leinert94})  &15:&\hskip-0.25cm Jayawardhana \&      \\
 4& Bernacca et~al. (\cite{Bernacca93})     &10& Leinert et~al. (\cite{Leinert97})  &  &\hskip-0.25cm    Brandeker (2001)    \\
 5& Bertout et~al. (\cite{Bertout99})      &\nix                     &16:&\hskip-0.25cm Rossiter (1955)        \\
 6& Brandner et~al. (\cite{Brandner95})     &12& Pirzkal et~al. (\cite{Pirzkal97})  &\nix                     \\
\end{tabular}\end{tiny}
\end{table}